vc/Week 32 starting 7 aug 06
SM18 Operation Meeting (Tuesdays) 08 Aug 06

Bench status  on Monday  07-08-2006 at: 8/7/2006 1:22:03 PM  
	LHC SM18 Magnet Test Facility 


	4
	A1
	MBBR2362
	PT 14 Magnet MAPS
	299.11

	12
	A2
	MBAR2333
	ICS 5 Traveller Last
	299.11

	10
	B1
	SSS314
	ICS 4 Final connection
	298.49

	2
	B2
	SSS528
	PT5.3: Powering Corrector CD
	2.35

	11
	C1
	SSS276
	PT 12 Warm Up
	237.58

	5
	C2
	SSS336
	PT 14 Magnet MAPS
	299.02

	6
	D1
	SSS646
	WP17 - HV Test
	297.41

	9
	D2
	SSSS618
	PT 12 Warm Up
	294.64

	1
	E1
	MBAR1303
	ICS 4 Final connection
	299.11

	7
	E2
	MBAL2281
	PT 13.2 Resist. Meas.
	298.04

	3
	F1
	MBAR1301
	PT 3 QH Measure
	2.35

	8
	F2
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	297.59


  Major WATER Station cleanout for 30 hrs starting Monday 7thAug 8 am
    
CONFIG :  D 6 - AS 3 -  IRSSS 3 ;   
Supervisor: marco  ;Eq support : 16-3602 
14 Magnet Tests completed  
6 Dipoles (), 
5 Arc-SSS  tests but ONLY 4  magnets( 312 ROBTC)
0  500-series [ ]
2  600-series [ 624 (165 hrs), 650 2nd run – MM done (169 hrs) ]
General Points :
E2 Bench will be  blocked by   2281 for nearly 10-12 days when it goes!! 

· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· News /Situation from (SSS, SSSS)  :  J Halik
· Situation SSSS  :D1 & D2 issues
· A.O.B..
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1 [TBF2 |VBBR1362 [Fake (28072006 |1 | 42h. | 25h. |22h.| ISh Bh 02082008 17 |2 swipping

2 |[TBFt |MBBRI364 |False |2507-2006 [t | 3th. | 20h. [19h.| 14h xh Broroos | 12 |3 | svppng

3 TBFL MBBR2366 |[Fake [31:07-2006 1| 24h. | 33h. |15h.| 18h soh (05082006 12 2| suipping

4 TBEL MBAR13SS |[Fake [26:07-2006 1| 23h. | 20h. |20h.| 32h wh o700 | 2|3 | sippng

5 TBD2 [S55624 Felse |2707-2006 |t | 30h. | 13h. |63h.| ¢h 4ah (02082006 ° |7 | stndey

6 TED1 [SS5650 False (25072006 [2| 20h. | 2th. |72h.| 16h. sth (oros2006| 2 |12|  stndey

7 [TBC2 555267 False 29072006 [t | 27h. | 32h. [26h.| 10h. 14h (03082006 | 2% |3 | swppng

8 [TBC1 [s55312 False [28:07-2006 |2 | 30h, Xh. |S4h.| 14h oh.  [02:08-2008 | 13 |g | MotFounder
h Thermal cyde

9 [TBC1 [sss312 False (02082006 [2 | oh. ith. [14h.| 6h 7h. (04082006 [48h. 2| Suppng

10[78B2 [ss5232 Fase [31072006 2| 33h. | 3h. [20h. | sh 15h. (04082006 [89h. 3|  Suppng

11[78B2 [s55259 False [2607-2006 [1 [ 20h. | 12h. [15h.[ 2h 15h.  [31072006 [96h. 3| Suppng

12[TBB1 [s55320 False [27072006 [t [ 22h. | 15h. [42h.[ 12h. 8h.  [3107-2006 [99h. [3 | Stppra

13[TBA2 |MBAR2361 |Fdse |26:07-2006 [1| 37h. | 37h. |26h.| 22h 2ah, 02082008 | ¥ |3 | svppng

14/TBAL [MBBR2362 |Fakse [29:07-2006 |2 | 25h 27h. |s4h.| 19h oh.  [0408-2008 | 125 |g | MotFoundor
h Thermal cyde
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650 ..was 2nd run here  & with MM after problems of anticryo & shafts in 1st run ; 624 had 8 quenches
312 had ROBTC
	01M (TUE)

	TBA1
	Taking long time to reach 1.9. Checked with CRYO team. They said that they are not sure what is the problem and are looking in to it.
	 (Software problem - control of valves)

	TBC
	Cluster C : Quench Heater 1 and 2 not charging. Interconn box not switching properly. Frederico informed. He will look in to the problem.
	QH charging problem solved by Federico. QH matrix card 1 was not fully inserted.

	TBE2
	TBE2 : It was observed that a cryo sensor cable was shorting the main magnet current lead on top of CFB. This was sorted out and HV test is re launched.
	

	“
	TBE2: The HV tests are failing again. waiting again for condensation to clear.
	

	TBE
	Cluster E: Hardware recognition failure. Federico informed.
	TT821 coeff data are missing again in MTF database


	01A (TUE)

	TBE2
	 HV failed again after provoked quench even with a different HV rack. So it is signed for warm up as decided.


	

	“
	MBAL2281 HV problem persists. Fredrico, in consultation with TC (Arjan) & Pear, decided to perform a provoke quench (1.5kA) & repeat CDBP to check the performance. PT3 was ok.
	








01 AUG 06 (TUE Morning) to 07 AUG 06 ( MON Night)
	01N(TUE)


	TBC1
	 LF acquisition didn't generate any file during provoked quench (for warming up the magnet for ROBTC), instead the message "Generic error occurred” 


	ES may please look into this as this is repeatedly coming.


	02M(WED)


	TBA1
	 The magnet on A1 can not be removed. Rocla access is blocked by 2 vehicles, an anticryostat rack, one SSW stage and black magnet end-caps.....
	


	02A(WED)


	TBA1
	Helium level fluctuation on current lead B, leading to intermittent loss of CRYO OK, resulting in automatic SPA in the 14KA main power supply. The automatic SPA occurred twice (at 11400 A and at 12000 A after 11 min). ACR informed. They asked to wait for 20 min and then try again.
	He consulted ACR team and found out that due to fluctuation in the temp of current lead B the cryo ok is lost while the test is in progress. ACR team will watch the temp. while the test is carried out next time.

	“
	 Third trial also failed. The magnet stayed at 12000 only for 5 minutes. Frederico was called. 
	

	TBE2
	Pierre told that the HV test at warm will have to be done at safe voltage levels, requiring new sequence to be defined and run. Therefore he insisted that Frederico will have to do this test.
	Frederico has done HV(warm) & reported that it has failed and asked to kept as it is and told that he will investigate tomorrow.

	A,B,C
	Cluster A,B,C : Magnet name missing in Tema. No tests could be done.
	Frederico called Alessandro Reimondo. He came and set right the problem

	TBB1
	Problem in Quadrupole Vs. ground step of HV test (CDBP). TC contacted. He in turn talked to Pierre. Pierre instructed OP team to carry out provoked quench at 1.5 KA and then repeat the HV test. The HV test, after provoke quench, have failed again. 
	


	02N(WED)


	TBB1
	On B1, PT2 failed at Quad Vs Gnd after 3kA Quench. 
	


	03M(THU)


	TBE2
	HV done manually by ES at 1000V with power cables removed: result is ok. HV @ warm to be repeated as soon as the power cables are reconnected this afternoon.
	

	TBF2
	Prep 5: Initial convalescence voltage was observed to be 3.17V instead of ~11.5V. This is not the case when P10 cable is removed 
	Frederico investigating the cable fault & cleared


	03A(THU)


	TBD2
	 New magnet S4-618 has mismatch between bar code label, alphanumeric label and name in MTF database. Bench configured with MTF database name 


	

	TBE2
	 Power cables connected. Warm HV (Prep 3) done as instructed by eqpt.support.
	

	“
	HV result OK. Frederico suggested to cool down the magnet again and launch cold HV. TC also recommended to cool down the magnet, and to consider it as not repeated one. 
	


	03N(THU)


	TBB1
	Eqpt. support contacted for HV problem on TBB1. Prep3 (HV at warm) was done Cryo was informed to cool down the magnet Since the E-Traveller was already signed till sL. No. 7.3. The cryo operator requested to un-sign the e traveller for cooling down. Hence two last signatures were unsigned. Further un-signing is not allowed by the E- Traveller. Cryo team is informed.


	Problems with going back in e-traveller


	04M(FRI)



	TBD2
	 Anticryostat end-plate yet to be connected on aperture 1. ICS has signed for warm tests. Complete scanning not possible as end plates are not put in place on the MRB side. IFS box seal not opened. Contacted ICS and asked them to make the IFS box accessible for carrying out warm tests. Halik informed that the connections are completed and electrical tests can continue.
	J Halik informed that some connections are yet to be made and then the end plates will be put. It was done later.




	04A(FRI)



	TBB1
	After warming up and cooling down, CQBP again failing at Quderpole Vs Gnd second step 900 Volt as voltage > 500 Volt. Bypassed this step and launched the test again, it cleared rest of all the steps. So inform to equipment support, they suggested to do 3.0 KA quench. But if the HV fails warm up.
	

	04N(FRI)


	TBA1
	 During PTC 6 instability in the cryo occured at current 12850A after 3 min hold, which resulted in automatic SPA trigger & current ramp down to zero. Informed to TC & TC suggested to repeat the test again.
	

	TBB1
	 (SSS 308): PT 2 again failed Quad Vs Gnd failed at 900V step. As per information given by the last shift magnet is put for warm up.
	


	05M(SAT)


	TBE2
	for 2281:- Cold HV (CDBP) failed again. TC instructed to perform a provoke quench @1.5kA (all HF) and repeat HV manual step. If PT2 fails again magnet can be warmed up.


	

	TBB1
	for 308:- Cold HV (CDBP) failed after 2nd attempt of cooldown. Tc asked to warm it up .
	


	06A(SUN)


	TBB2
	 Unable to launge the correctors tests CD & EF => PC Aux fault / ES and PO piquet called : PC fault or /and regulation problem 
	

	“
	With the PO piquet, PLCSC was reset a 2ndtime.'PC Aux' error disepeared. 
	Done on 7th morning.


	07M(MON)   Water supply problem : No cold tests could be done



	07A(MON)


	TBE2
	2281: TC (M Buzio) said that he will ask the ICS to proceed with disconnection of magnet so that repair work can be done off the bench. 
	

































































